Author's response to reviews

Title: Smoking Status Impacts microRNA Mediated Prognosis and Lung Adenocarcinoma Biology

Authors:

Emily A Vucic (evucic@bccrc.ca)
Kelsie L. Thu (kthu@bccrc.ca)
Larissa A. Pikor (lpikor@bccrc.ca)
Katey S.S. Enfield (kenfield@bccrc.ca)
John Yee (John.Yee@vch.ca)
John C. English (John.English@vch.ca)
Calum E. MacAulay (cmaaulay@bccrc.ca)
Stephen Lam (slam2@bccancer.bc.ca)
Igor Jurisica (juris@ai.utoronto.ca)
Wan L. Lam (wanlam@bccrc.ca)

Version: 5 Date: 23 September 2014

Author's response to reviews: see over
September 23, 2014

RE: Revision#2 for “Smoking Status Impacts microRNA Mediated Prognosis and Lung Adenocarcinoma Biology”
MS#: 1554309953130264

To: Ms. Cherry Battad
Journal Editorial Office
BioMed Central

Dear Ms. Battad,

Please find enclosed our revised manuscript. We have had the manuscript, tables, figure legends and supplementary files thoroughly edited for clarity and grammar. We ensure the enclosed manuscript and files conform to BMC Cancer’s journal style and formatting requirements.

Thank you very much for your efforts in helping us to improve our manuscript. We look forward to hearing from you.

Sincerely,
Emily Vucic